Author’s response to reviews

Title: UTERINE AND PLACENTAL BLOOD FLOW INDEXES AND ANTINUCLEAR AUTOANTIBODIES IN UNEXPLAINED RECURRENT PREGNANCY LOSS: SHOULD THEY BE INVESTIGATED IN PREGNANCY AS CORRELATED POTENTIAL FACTORS? A RETROSPECTIVE STUDY

Authors:

Valentina Bruno (valentinabruno_86@hotmail.it)
Carlo Ticconi (ticconi@uniroma2.it)
Federica Martelli (martelli.federica@gmail.com)
Marzia Nuccetelli (marzianuccetelli@yahoo.com)
Maria Capogna (maria.vittoria.cap77@gmail.com)
Roberto Sorge (sorge@uniroma2.it)
Emilio Piccione (piccione@med.uniroma2.it)
Adalgisa Pietropolli (pietropolli@med.uniroma2.it)

Version: 3 Date: 05 Jan 2020

Author’s response to reviews:

1. Please ensure the author list on the title page matches the list of authors on the authorship change form. If the list of authors has changed again, please submit an updated authorship change form.

2. Please include the full name of the ethics committee (and the institute to which it belongs to) that approved the study and the committee’s reference number if appropriate.

3. If abbreviations are used in the text they should be defined in the text at first use, and a list of abbreviations should be provided in the Declarations.

4. The specific individual contributions of ALL authors to the manuscript should be specified in the Authors’ Contributions section. Guidance and criteria for authorship can be found here: http://www.biomedcentral.com/submissions/editorial-policies#authorship

5. Please remove the answer to reviewers document from the end of your manuscript file as this is no longer needed in the publication process.
6. At this stage, please upload your manuscript as a single, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Answer: We have now gone through all the 6 points checklist and made the appropriate corrections, as suggested.

Reviewer reports:

Jean Christophe Gris (Reviewer 1): Please include all comments for the authors in this box rather than uploading your report as an attachment. Please only upload as attachments annotated versions of manuscripts, graphs, supporting materials or other aspects of your report which cannot be included in a text format. Please overwrite this text when adding your comments to the authors.

Answer: We have now uploaded all the attachments files, as requested.

This revised version of the text has been improved to some extent. I have no worrying remarks to make.

Answer: We would like to thank you to make the manuscript improved with your challenging suggestions.